BROOKHAVEN NATIONAL LABORATORYASSOCIATED UNIVERSITIES, INC.Upton. Long Island, New York 11 973-5000TELEX: 6852516BITNET:RHIC@BNLDAG FAX: 516-282-2588TOTAL P.02 (516) 282 7753 FTS 666 7753RHIC Project Bitnet - ludlam@bnldag FAX - (516) 282-258818 September 1992Dr. Jay MarxSTAR Project DirectorLawrence Berkeley Laboratory1 Cyclotron RoadBerkeley, CA 94720Dear Jay,At the TAC meeting you raised the question of whether the use of a conventional magnet forSTAR would cause the operation of the experiment to be curtailed during times of peak power demand atthe Laboratory.The most recent estimate of the power required for a conventional solenoid for STAR is 3.1 MW(including cooling tower, chiller, etc.), giving a total power requirement of about 3.6 MW for theexperiment. The Laboratory is planning that the total peak power consumption for RHIC (includinginjectors, the collider, experiments and building power) will be about 46 MW, with a 40 week annualrunning period. We estimate a total site load of about 80 MW when RHIC becomes operational. Of theRHIC total, about 9 MW has been budgeted for the experiments during normal machine operation (5MW during machine cooldown). Of the other large experiments now being considered, PHENIX and theForward Hadron Spectrometer are each expected to consume about 1 Megawatt of power, with PHOBOSrequiring much less. Thus the operation of STAR with a conventional magnet will still leave acomfortable margin in the power available for the experimental program.Yours sincerely,Thomas W. LudlamAssoc. Project HeadDetectors & Exp. SupportTWL:bescc: S. Ozaki.